Author's response to reviews

Title: The Diagnostic Value of Ultrasonography in Carpal Tunnel Syndrome: A Comparison between Diabetic and Non-Diabetic Patients

Authors:

Nai-Wen Tsai (tsainw@yahoo.com.tw)
Lian-Hui Lee (napaj@adm.cgmh.org.tw)
Chi-Ren Huang (suika68@adm.cgmh.org.tw)
Wen-Neng Chang (cwenneng@ms19.hinet.net)
Hung-Chen Wang (m82whc@yahoo.com.tw)
Yu-Jun Lin (lin-yu-jun@yahoo.com.tw)
Wei-Che Lin (u64lin@yahoo.com.tw)
Tsu-Kung Lin (tklin@adm.cgmh.org.tw)
Ben-Chung Cheng (benzmcl@yahoo.com.tw)
Yu-Jih Su (bensu8@gmail.com)
Chia-Te Kung (kungchiate@gmail.com)
Su-Fong Chen (fangoe@adm.cgmh.org.tw)
Cheng-Hsien Lu (chlu99@ms44.url.com.tw)

Version: 2 Date: 12 April 2012

Author's response to reviews: see over
Dear editors:

We are apologized for uploading a wrong cover letter file in the previous version. We did not submit the manuscript to other journal previously. We are actually submitting the manuscript entitled: “The Diagnostic Value of Ultrasonography in Carpal Tunnel Syndrome: A Comparison between Diabetic and Non-Diabetic Patients” for consideration of publication on *BMC neurology*.

All authors have made a substantial contribution to this manuscript, have read and approved the final manuscript, and have no substantial direct or indirect commercial financial incentive associated with publishing this article. In this study, Informed consent was obtained from all of the study subjects. The study was approved by the hospital’s Institutional Review Committee on Human Research. We declare that this manuscript or portions are not under consideration by another journal and have not been previously published and will not be submitted elsewhere if accepted by the journal and not before the Editorial Board has reached a decision.

Thank you very much for your consideration!

Sincerely

Corresponding author: Cheng-Hsien Lu, MD

Department of Neurology, Chang Gung Memorial Hospital

123, Ta Pei Road, Niao Sung Hsiang, Kaohsiung Hsien, Taiwan

TEL: +886-7-7317123 ext 2283    FAX: +886-7-7902684

E-mail: chlu99@ms44.url.com.tw

April 12, 2012